Imatinib in gastrointestinal stromal tumour: Northern Cancer Network experience

F Azribi¹, ARA Razak², P Dildey³, J Adam⁴, J Wilsdon⁴ and M Verrill

¹Northern Institute for Cancer Research, Paul O'Gorman Building, University of Newcastle, Newcastle, NE2 4HH, UK

²Drug Development Programme, Suite 702, 7th Floor, Princess Margaret Hospital, 610 University Avenue, Toronto, ON M5G 2M9, Canada

³Royal Victoria Infirmary, Queen Victoria Road, Newcastle, NE1 4LP, UK

⁴Northern Centre for Cancer Care, Freeman Hospital, Freeman Road, Newcastle, NE7 7DN, UK

Abstract

Imatinib treatment in metastatic or inoperable gastrointestinal stromal tumours (GIST) has shifted the paradigm of treatment of this disease. Successful clinical trials of imatinib led to rapid regulatory approval and, in England and Wales, National Institute for Health and Clinical Excellence (NICE) guidance on use of this technology. NICE recommend detailed audit of their guidelines in clinical practice. This audit reflects that guidance and was designed to document the use of imatinib in routine clinical practice.

Methods: We conducted a retrospective audit of patients with GIST treated with imatinib from 1 February 2002 to 31 March 2007. Information gathered included patient demographics, disease characteristics and details of treatment administered, treatment response, toxicities and follow-up data. The primary objective was to record disease control rate (DCR), defined as a lack of progression on computed tomography at three months. Secondary end points of this audit were progression-free and overall survival. These were compared with published clinical trial results.

Results: Thirty-six consecutive patients with a diagnosis of GIST treated with imatinib were identified. Median age of patients was 70.1 years. At the time of analysis, patients have been followed up for a median of 41.6 months. In total, patients were treated for a median of 15.8 months. Treatment was generally well tolerated with a small percentage of patients experiencing grade 3/4 toxicities. Disease control was observed in 30 patients (DCR, 83.3%, 95% CI 67.2–93.6, intention to treat analysis). The median progression free survival (PFS) in this cohort was 23.7 months (95% CI 12.9–34.4); while the median overall survival was 39.7 months (95% CI 22.8–56.5).

Conclusion: Our data demonstrated that the treatment of unselected GIST patients within the NICE guidance compares favourably to previously published data of randomized registration studies of imatinib. Of note, the median age of this cohort is some ten years older than that reported in the trials. Imatinib was well tolerated with acceptable treatment-related adverse events.

Published: 14/12/2009

Received: 10/10/2009

ecancer 2009, 3:162 DOI: 10.3332/ecancer.2010.162

Copyright: © the authors; licensee ecancermedicalscience. This is an Open Access article distributed under the terms of the Creative Commons Attribution License (<u>http://creativecommons.org/licenses/by/2.0</u>), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

Competing Interests: The authors have declared that no competing interests exist.

Correspondence to M Verrill. Email: Mark.Verrill@nuth.nhs.uk

Introduction

The term gastrointestinal stromal tumours (GIST) was first coined by Mazur and Clark to describe gastrointestinal nonepithelial neoplasms that had neither smooth muscle nor Schwann-cell features [1]. They are thought to be derived from the interstitial cells of Cajal, which are the pacemaker cells of gastrointestinal tract [2].

Although GISTs are the commonest mesenchymal tumour of the gastrointestinal tract, they are relatively rare. Its exact incidence remains unknown, but is estimated at 14.5 cases per million in Sweden [3]. They commonly occur in people above the age of 50, with almost a similar distribution across the male and female gender [3,4]. GISTs are most commonly found in the stomach and small bowel, but they can occur in all other parts of the gastrointestinal tract as well as the abdominal and pelvic cavity. Liver and peritoneum are the most common sites of metastases [5]. The malignant potential of GIST depends on some features of the tumour, mainly size and mitotic rate [6]. The larger the size and the higher the mitotic rate, the more aggressive behaviour and risk of relapse the GIST will have. The location of GIST also has an impact on the outcome, with gastric tumours having a better prognosis than the intestinal tumours of the same features [7].

Definite diagnosis of this tumour requires histological confirmation. The landmark feature is the positive staining for CD117 receptor, also known as c-kit receptor. It is a transmembrane tyrosine kinase receptor for the cytokine stem cell factor. The majority of GISTs (circa 85%) expresses this protein. GISTs are often also positive for CD34, a cell surface protein also found in blood progenitor cells and endothelial cells. Less frequently, GIST cells are positive for smooth muscle actin (SMA), and they are rarely positive for the marker of cells derived from the neural crest (S100) [2,6,8,9].

Most GISTs have a gain of function mutation in the *KIT* protooncogene (*cKIT*) that translates into a constitutive ligand independent activation of the tyrosine kinase receptor [10]. This is thought to be the key event in the pathogenesis of GIST [11]. However, a minority of GISTs contain mutations in the homologous kinase platelet-derived growth factor receptor alpha (PDGFR- α) gene [12]. Mutations of these two closely related tyrosine kinases are mutually exclusive; with about 85– 90% of GISTs, having a mutation in one of these two kinase genes [13]. The remaining GISTs carry the wild *KIT* gene type. To date, there are several mutations identified that have therapeutic implications. The most common *cKIT* mutation, exon 11, shows good response to the tyrosine kinase inhibitor imatinib and better overall survival (OS) than the less commonly occurring exon 9 mutation and the wild type GIST [14].

Surgery is the mainstay treatment for small-localized tumours. For locally advanced, inoperable or metastatic disease, tyrosine kinase inhibitor, imatinib mesylate, is the current worldwide standard first line treatment. It represents a revolution in the treatment of this notoriously chemotherapy-resistant tumour. Historically, the response rate to chemotherapy was very low, and the median survival for advanced disease was between 12 and 24 months [4,5,15]. Several phase 1 and 2 trials established the high efficacy and good tolerability of imatinib in treatment of GISTs [16-18]. A pivotal randomized trial conducted mainly in Europe showed an overall response rate of 52% and stable disease in 32% of patients [19]. The findings of the above trial was confirmed by another large trial, demonstrating an overall response rate of 45% while stable disease was achieved in 25% of patients [20]. With the use of imatinib, the median overall survival has improved dramatically, with a median overall survival exceeding 50 months [20,21]. The treatment protocol for patients in England and Wales with GIST is mandated by the National Institute for Health and Clinical Excellence (NICE) guidance (www.nice.org.uk/TA86/ guidance). In essence, this guidance recommended: (i) treatment with imatinib for patients with unresectable/metastatic GIST at 400 mg daily under the supervision of cancer specialists with experience in the management of this disease; (ii) patients should be assessed on a 12-weekly basis and kept in treatment if the disease is responding to treatment, defined as stable disease or better; and (iv) patients who have progressed on imatinib therapy should have their treatment discontinued. NICE also recommended auditing the above guidance against local practices.

Despite the impressive results with imatinib treatment of this disease, little is known with regards to its efficacy in day-to-day clinical setting. We therefore undertook a retrospective audit of consecutive, unselected patients with GISTs who were treated with imatinib within the NICE guidance.

Methods

Patients

Potential patients were identified from the Northern Cancer Network (NCN) soft tissue and bone sarcoma multidisciplinary meeting (MDM) registry from 1 February 2002 to 31 March 2007. The NCN covers a population base of more than two million people and is responsible for oncological care for patients from the north-east of England. The NCN soft tissue and bone sarcoma MDM comprises dedicated pathologists, radiologists, surgeons and oncologists.

Patients with a histological confirmation of GIST who were treated with imatinib were included in this series. We examined patient demographics, disease characteristics and details of treatment administered. Information on treatment response, treatment failure, toxicities and follow-up were also collected. Each patient record was examined and data were retrieved for the above parameters. This audit was approved by the Caldicott Committee of the Newcastle upon Tyne Hospitals NHS Foundation Trust, which acts as the custodian of patient records and confidentiality at our institution.

Treatment

Patients were treated as per recommendation of the NICE guidelines. All patients were treated with imatinib mesylate, 400 mg once daily p.o. unless specified otherwise. Patients were reviewed weekly for the first four weeks, then monthly for two-month and three-month visits from thereon. All patients had computed tomography (CT) evaluation pre-treatment, with three-month CT scans planned throughout treatment duration. More recently, a follow-up schedule with fewer visits has also been adopted.

Audit end points

The primary end point of this retrospective audit was visual radiographic assessment for response and the clinical impression. This was evaluated as disease control rate (DCR), defined as a lack of clinically relevant progression on computed tomography (CT) at three months post-treatment initiation. We chose DCR as our primary end point as it reflects everyday clinical practice and it reflects accordance to the NICE guidelines. Whilst we also collated data on conventional response parameters (complete response (CR), partial response (PR), stable disease (SD) and disease progression (PD)), we acknowledge the limitations of response assessment, using conventional response parameters such as RECIST parameters in this disease [22].

Secondary end points of this audit included progression-free and overall survival. Progression-free survival (PFS) was defined as the time from start of imatinib treatment until documented progression or death from any cause. Overall survival was defined as the time from start of imatinib treatment to the time of the last review or death from any cause. Patients who were alive on the date of last follow-up were censored on both analyses.

Safety profile

Common toxicity criteria according to the National Cancer Institute Grading System (version 3.0) were used in the evaluation of treatment toxicity.

Statistical methods

The Kaplan-Meier method was used to generate estimation of progression-free and overall survival [23].

Results

Patient characteristics and treatment details

From 1 February 2002 to 31 March 2007, 36 patients with a diagnosis of GIST underwent treatment with imatinib within the Northern Cancer Network. Baseline characteristics of patients, their disease and previous treatment are summarized in <u>Table 1</u>. At the start of treatment with imatinib, 27 patients (75%) had metastatic disease, whilst five patients (13.9%) had locally advanced inoperable disease. Four patients (11.1%) were treated with imatinib despite having localized disease only as they were medically unfit for surgery.

In total, patients were treated for a median of 15.8 months. Two patients started imatinib at 200 mg daily p.o. as a precaution against bleeding problems. No patients had a dose of imatinib more than 400 mg o.d. Treatment deferral was documented in five patients due to toxicity. At the time of analysis, 26 out of 36 patients have had their imatinib treatment discontinued. Primary reason for treatment discontinuation was progressive disease (PD) (n=20), followed by treatment toxicity (n=6). All patients with progressive disease had their treatment discontinued.

Treatment response and benefit

Thirty-four patients were evaluable for treatment response. Two patients (5.6%) were not evaluable for radiological response due to early clinical deterioration. DCR was observed in 30 patients (83.3 %, 95% CI 67.2–93.6, intention to treat analysis). There were no CR, whilst progressive disease was noted in four patients (11.1%). Treatment failure denoted by progressive disease on radiological assessment or the occurrence of early clinical deterioration occurred in six patients (16.7%).

Table 1: Patient, disease and previous treatment demographics characteristics of patients, disease and previous treatment

Details	n	%
Age		
Median (range)	70 1 /27 96	
Gender	70.1 (3786)	
Male		
	17	47.2
Female	10	50.0
	19	52.8
Location of primary disease		
Stomach Small bowel	19	52.8
Rectum	4	11.1
	3	8.3
Others/Unknown*	10	27.8
Primary presenting symptom	10	27.0
Gastrointestinal bleeding	44	20.0
	11	30.6
Anaemia	6	16.7
Abdominal pain		
Abdominal mass	4	11.1
Other	3	8.3
	3	8.3
Asymptomatic	9	25.0
	9	25.0
CD117 staining Positive		
	35	97.2
Negative/Not done [#]	1	2.8
Previous surgery	I	2.0
Yes		
No	18	50
	18	50
Median (range)		
	1 (14)	
Intent at primary surgery (<i>n</i> =18)		
Curative		
Palliative		
	15	83.3
	3	16.7
	5	10.7

*Five located within the abdomen, whilst another five located within the pelvis.

[#]Pathological diagnosis could not be achieved despite repeated attempts. Treatment decision was based on clinical findings and recommendation of the MDM.

The median PFS in this cohort was 23.7 months (95% CI 12.9–34.4), whilst the median OS was 39.7 (95% CI 22.8–56.5). The PFS and OS actuarial plot is represented by <u>Figures 1</u> and <u>2</u>, respectively.

Toxicity

In general, treatment with imatinib was tolerated well. There were no grade 3/4 haematological toxicities necessitating



Figure 1: Progression-free survival curve—Kaplan-Meier plot for progression-free survival for all patients.

treatment deferral in our series. In 14 patients (38.8%), treatment was tolerated with no clinical toxicity. Non-haematological toxicities, when they occurred, were usually mild, with periorbital oedema (n=9), nausea (n=4), diarrhoea (n=3) and rash (n=3) being the commonest complain. However, a small proportion of patients developed grade 3/4 non-haematological toxicities and these are documented in Table 2.

Discussion

The use of imatinib in the management of inoperable and metastatic GIST represented a sea change in the management of these tumours. For many years, GISTs were not recognized as a separate clinical entity from leiomyosarcomas [24], and it was only really when the opportunity for targeted therapy using imatinib in this disease was mooted that clinicians began to treat them as a separate clinical identity from other retroperitoneal sarcomas.

Following on from the success of imatinib in the management of chronic myeloid leukaemia [25], the role of this drug in GISTs was then established with a series of rapidly accruing phase 1, 2 and 3 trials conducted in parallel in Europe and North America, leading to the rapid licensing of the drug followed by approval in the United Kingdom from NICE.

One of the inevitable consequences of this rapid regulatory approval was that clinicians treating the disease knew the sometimes dramatic effect of imatinib treatment, but very few had experience of using the drug in real life. One distinct clinical group, the extreme elderly, who many would have not considered clinical trial material and their treatment of GIST

Table 2: Grade 3/4 non-haematological toxicities

Details	n	%
Nausea/vomiting	3	8.3
Cardiac	2	5.6
Rash	1	2.8
GI bleeding	1	2.8





patients on imatinib is response assessment. As alluded to earlier, we have chosen DCR as the primary end point for response evaluation to imatinib as opposed to traditional end points such as RECIST as it may be inaccurate in this setting. However, at present, the use of positron emission tomography imaging is fast becoming the standard modality for response evaluation for this disease in many institutions, negating the shortcomings of CT response assessment [26,27].

The NCN soft tissue and bone sarcoma group took an early lead in making imatinib available for patients in the region and the contribution of sarcoma, gastrointestinal (both upper and lower) and hepatobiliary surgeons to regular meetings along with the presence of specialist sarcoma and gastrointestinal pathologists, radiologists and oncologists undoubtedly contributed to the ability to manage these patients effectively.

This series represents the results of multi-disciplinary management of this group of patients. The management guidelines were, essentially, fixed by the NICE guidance on the management of GIST, which is applicable locally and has been followed closely. This has enabled the generation of detailed individual patient data on response and treatment outcome. Our response rate (to include patients who had partial response and stable disease), time to progression and overall survival map almost exactly onto the published trial data from both the EORTC and the American Intergroup studies and show that in a multi-disciplinary setting in a UK centre, it is possible to match the results of clinical trials in the population as a whole, not just in those patients who would be eligible for clinical trials. Our results underpin both the efficacy and safety of imatinib treatment despite the high median age of patients in the cohort. This age distribution makes our experience unique among the published series.

Our unexpected finding of cardiac toxicity in two of the patients in the series highlights the importance of continuing pharmacovigilance after product licensing. This was not a finding that was expected from the registration studies and may, in part, be a reflection of the age group treated. It raises the question of a need to monitor cardiac function in selected patients although our review does not have the power to answer this.

Conclusion

Our data show that in routine clinical practice based on a comprehensive multi-disciplinary team framework, the impressive results produced by imatinib in GISTs in the randomized pivotal trials can be reproduced. The treatment can be delivered safely and the response rates and other clinical outcomes map onto the published clinical trial data. This audit meets the NICE requirement for practice evaluation and our results confirm the appropriateness of the NICE guidance to UK practice.

References

- Mazur MT and Clark HB (1983) Gastric stromal tumors. Reappraisal of histogenesis Am J Surg Pathol 7 507–19 PMID: 6625048 10.1097/00000478-198309000-00001
- Kindblom LG, Remotti HE, Aldenborg F and Meis-Kindblom JM (1998) Gastrointestinal pacemaker cell tumor (GIPACT): gastrointestinal stromal tumors show phenotypic characteristics of the interstitial cells of Cajal Am J Pathol 152 1259–69 PMID: 9588894
- Nilsson B, Bümming P, Meis-Kindblom JM, Odón A, Dortok A, Gustavsson B, Sablinska K and Kindblom LG (2005) Gastrointestinal stromal tumors: the incidence, prevalence, clinical course, and prognostication in the preimatinib mesylate era—a population-based study in western Sweden Cancer 103 821–9 PMID: 15648083 10.1002/cncr.20862
- Gold JS, van der Zwan SM, Gönen M, Maki RG, Singer S, Brennan MF, Antonescu CR and De Matteo RP (2007) Outcome of metastatic GIST in the era before tyrosine kinase inhibitors Ann Surg Oncol 14 134–42 PMID: <u>17080234</u> 10.1245/s10434-006-9177-7
- DeMatteo RP, Lewis JJ, Leung D, Mudan SS, Woodruff JM and Brennan MF (2000) Two hundred gastrointestinal stromal tumors: recurrence patterns and prognostic factors for survival Ann Surg 231 51–8 PMID: 10636102 10.1097/00000658-200001000-00008
- Fletcher CD, Berman JJ, Corless C, Gorstein F, Lasota J, Longley BJ *et al* (2002) Diagnosis of gastrointestinal stromal tumors: A consensus approach *Hum Pathol* 33 459–65 PMID: 12094370 10.1053/hupa.2002.123545
- Miettinen M and Lasota J (2006) Gastrointestinal stromal tumors: pathology and prognosis at different sites Semin Diagn Pathol 23 70–83 PMID: 17193820 10.1053/ j.semdp.2006.09.001
- Miettinen M and Lasota J (2001) Gastrointestinal stromal tumors—definition, clinical, histological, immunohistochemical, and molecular genetic features and differential diagnosis Virchows Arch 438 1–12 <u>PMID:</u> <u>11213830</u> 10.1007/s004280000338
- Sarlomo-Rikala M, Kovatich AJ, Barusevicius A and Miettinen M (1998) CD117: a sensitive marker for gastrointestinal stromal tumors that is more specific than CD34 Mod Pathol 11 728–34 PMID: 9720500
- Hirota S, Isozaki K, Moriyama Y, Hashimoto K, Nishida T, Ishiguro S *et al* (1998) Gain-of-function mutations of ckit in human gastrointestinal stromal tumors *Science* 279 577–80 PMID: 9438854 10.1126/science.279.5350. 577
- 11. Rubin BP, Singer S, Tsao C, Duensing A, Lux ML, Ruiz R et al (2001) KIT activation is a ubiquitous feature of

gastrointestinal stromal tumors Cancer Res 61 8118–21 PMID: 11719439

- Corless CL, Schroeder A, Griffith D, Town A, McGreevey L, Harrell P, Shiraga S, Bainbridge T, Morich J and Heinrich MC (2005) PDGFRA mutations in gastrointestinal stromal tumors: frequency, spectrum and in vitro sensitivity to imatinib J Clin Oncol 23 5357–64 PMID: 15928335 10.1200/JCO.2005.14.068
- 13. Rubin BP, Heinrich MC and Corless CL (2007) Gastrointestinal stromal tumour Lancet 369 1731–41 PMID: 17512858 10.1016/S0140-6736(07)60780-6
- Badalamenti G, Rodolico V, Fulfaro F, Cascio S, Cipolla C, Cicero G et al (2007) Gastrointestinal stromal tumors (GISTs): focus on histopathological diagnosis and biomolecular features Ann Oncol 18 136–40 PMID: <u>17071931</u> 10.1093/annonc/mdm243
- Ng EH, Pollock RE, Munsell MF, Atkinson EN and Romsdahl MM (1992) Prognostic factors influencing survival in gastrointestinal leiomyosarcomas. Implications for surgical management and staging Ann Surg 215 68–77 PMID: 1731651 10.1097/00000658-199201000-00010
- van Oosterom AT, Judson I, Verweij J, Stroobants S, Donato di Paola E, Dimitrijevic S *et al* (2001) Safety and efficacy of imatinib (STI571) in metastatic gastrointestinal stromal tumours: a phase I study *Lancet* 358 1421–3 <u>PMID: 11705489</u> <u>10.1016/S0140-6736(01)06535-7</u>
- 17. Verweij J, van Oosterom A, Blay JY, Judson I, Rodenhuis S, van der Graaf W *et al* (2003) Imatinib mesylate (STI-571 Glivec, Gleevec) is an active agent for gastrointestinal stromal tumours, but does not yield responses in other soft-tissue sarcomas that are unselected for a molecular target. Results from an EORTC Soft Tissue and Bone Sarcoma Group phase II study *Eur J Cancer* **39** 2006–11 <u>PMID: 12957454</u> <u>10.1016/S0959-8049(02)00836-5</u>
- Demetri GD, von Mehren M, Blanke CD, Van den Abbeele AD, Eisenberg B, Roberts PJ et al (2002) Efficacy and safety of imatinib mesylate in advanced gastrointestinal stromal tumors N Engl J Med 347 472– 80 PMID: 12181401 10.1056/NEJMoa020461
- Verweij J, Casali PG, Zalcberg J, LeCesne A, Reichardt P, Blay JY et al (2004) Progression-free survival in gastrointestinal stromal tumours with high-dose imatinib: randomised trial Lancet 364 1127–34 PMID: <u>15451219</u> 10.1016/S0140-6736(04)17098-0
- 20. Blanke CD, Rankin C, Demetri GD, Ryan CW, von Mehren M, Benjamin RS *et al* (2008) **Phase III randomized, intergroup trial assessing imatinib mesylate at two dose levels in patients with unresectable or metastatic**

gastrointestinal stromal tumors expressing the kit receptor tyrosine kinase: S0033 *J Clin Oncol* 26 626–32 PMID: 18235122 10.1200/JCO.2007.13.4452

- Blanke CD, Demetri GD, von Mehren M, Heinrich MC, Eisenberg B, Fletcher JA *et al* (2008) Long-term results from a randomized phase II trial of standard- versus higher-dose imatinib mesylate for patients with unresectable or metastatic gastrointestinal stromal tumors expressing KIT J Clin Oncol 26 620–5 PMID: 18235121 10.1200/JCO.2007.13.4403
- Benjamin RS, Choi H, Macapinlac HA, Burgess MA, Patel SR, Chen LL, Podoloff DA and Charnsangavej C (2007) We should desist using RECIST, at least in GIST J Clin Oncol 25 1760–4 PMID: 17470866 10.1200/JCO.2006. 07.3411
- Kaplan EL and Meier P (1958) Nonparametric estimation from incomplete observations J Am Stat Assoc 53 457– 81 10.2307/2281868
- 24. Plaat BE, Hollema H, Molenaar WM, Torn Broers GH, Pijpe J, Mastik MF, Hoekstra HJ, van den Berg E, Scheper RJ and van der Graaf WT (2000) **Soft tissue**

leiomyosarcomas and malignant gastrointestinal stromal tumors: differences in clinical outcome and expression of multidrug resistance proteins *J Clin Oncol* **18** 3211–20 <u>PMID: 10986053</u>

- 25. O'Brien SG, Guilhot F, Larson RA, Gathmann I, Baccarani M, Cervantes F et al (2003) Imatinib compared with interferon and low-dose cytarabine for newly diagnosed chronic-phase chronic myeloid leukem N Engl J Med 348 994–1004 PMID: 12637609 10.1056/ NEJMoa022457
- McAuliffe JC, Hunt KK, Lazar AJ, Choi H, Qiao W, Thall P, Pollock RE, Benjamin RS and Trent JC (2009) A randomized, phase II study of preoperative plus postoperative imatinib in GIST: evidence of rapid radiographic response and temporal induction of tumor cell apoptosis Ann Surg Oncol 16 4 910–9 PMID: 18953611 10.1245/s10434-008-0177-7
- 27. Shinto A, Nair N, Dutt A and Baghel NS (2008) Early response assessment in gastrointestinal stromal tumors with FDG PET scan 24 hours after a single dose of imatinib *Clin Nucl Med* 33 7 486–7 <u>PMID: 18580237</u> 10.1097/RLU.0b013e31817792a3